Author's response to reviews

Title: Circulating Immune Complexes and trace elements (Copper, Iron and Selenium) as markers in oral precancer and cancer: A randomised controlled clinical trial

Authors:

Sunali S Khanna (sunali3011@yahoo.com)
Freny R Karjodkar (fkarjodkar@yahoo.co.in)

Version: 8 Date: 10 October 2006

Author's response to reviews:

Dear Editor,

The following sections have been changed in the manuscript. The abstract, title page, materials and methods, background, figures, legends, references, acknowledgements and formatting have been revised as per your requirements. Kindly accept and do the needful at the earliest.

Thanking you,

With fond regards,

Dr. Sunali Khanna